Author’s response to reviews

Title: Reliability and validity of ADHD diagnostic criteria in the Assessment System for Individuals with ADHD (ASIA), a Japanese semi-structured diagnostic interview

Authors:

Toshinobu Takeda (t-tak@umin.ac.jp)
Yui Tsuji (pyhlg_scooy@yahoo.co.jp)
Teruhisa Uwatoko (uwatoko.teruhisa.8s@kyoto-u.ac.jp)
Hiroshi Kurita (hkurita@mvf.biglobe.ne.jp)

Version: 2
Date: 5 February 2015

Author’s response to reviews:

Dear Dr. Teicher;

According to the editorial comments, we added the following information to the manuscript. We are sorry for the inconvenience.

1. E-mail addresses of all authors
2. Tables and a figure in the manuscript
3. Line numbering

Sincerely,
Tosinobu Takeda